Author's response to reviews

Title: Prevention of non-communicable disease in a population in nutrition transition: Tehran Lipid and Glucose Study Phase II (Study Protocols & Design)

Authors:

Fereidoun Azizi (Azizi@endocrine.ac.ir)
Arash Ghanbarian (ghanbarian@endocrine.ac.ir)
Farzad Hadaegh (fzhadaegh@endocrine.ac.ir)
Amir Abbas Momenan (momenan@endocrine.ac.ir)
Parvin Mirmiran (mirmiran@endocrine.ac.ir)
Mehdi Hedayati (hedayati@endocrine.ac.ir)
Yadollah Mehrabi (ymehrabi@gmail.com)
Saleh Zahedi-Asl (zahedi@endocrine.ac.ir)

Version: 2 Date: 28 May 2008

Author's response to reviews: see over
Dear Editor-in-chief

The manuscript was revised according to the comments of respected referees. All the comments were accepted and the following corrections were made in the text:

1. The manuscript was reformatted as "study protocol" and the title was partially changed.
2. The sampling method was corrected and more details were added.
3. Typos were corrected.
4. A table including "age distribution of Tehran, Iran, and study population" was added.
5. More details were added to research methodology and interventions.
6. "Authors' contributions" item was added and explained.
7. Table 1 and its relevant sentences were moved to "introduction". However, the authors believe the table can explain the studies well.
8. References 1 and 2 were deleted and the relevant sentences in the text were omitted.
9. The authors chose JNC-VI because in that time that they designed the study there was no JNC-VII.

Yours sincerely,

Fereidoun Azizi, M.D.

Professor of Internal Medicine and Endocrinology
Director of Research Institute for Endocrine Sciences,
Shaheed Beheshti University of Medical Sciences,
P.O. Box: 4763
Tehran 19395, Iran
Phone: +98 21 2243 2500
Direct: +98 21 2240 9309
Fax: +98 21 2240 2463 or +98 21 2241 6264
Email: azizi@endocrine.ac.ir